To the Editorial Board of the World Journal of Radiology & BPG:

On behalf of my manuscript’s team, I thank you for your continued support of our manuscript’s revision. After a significant month of discussion and revision. Our team is submitting our most revised manuscript for potential publication. We believe it has undergone significant grammatical revision by Native English speakers. In addition, we have updated our discussion with most recent citations to further improve our scientific validity. Below are the specific addressed comments made by the reviewers. We appreciate their time with our prior manuscript.

Reviewer Name: Anonymous
Review Date: 2022-03-13 16:10
Specific Comments To Authors: The systematic review is the first but the data pertaining to this area is limited with many procedural limitations. Secondly the subject needs robust study before any conclusions can be drawn. In addition, the following needs to be done for enhancing the manuscript: 1. Grammar corrections throughout 2. Please add this to the discussion Liang, B., Liang, Y., Li, R. et al. Effect of renal denervation on long-term outcomes in patients with resistant hypertension. Cardiovasc Diabetol 20, 117 (2021). https://doi.org/10.1186/s12933-021-01309-3

Authors: Thank you for this wonderful analysis of our manuscript. We agree that this is the first systematic review of case reports regarding this interventional radiology topic. Our motivation behind creating this study design has been included in our revised discussion and conclusion. Additionally, we have undergone significant grammatical revision with aid of native English speakers. Finally, we have included the source paper by Liang et al. 2021 which you have suggested. We believe the most pertinent reflection from this paper is its analysis of using office-based blood pressure readings to evaluate our outcomes. In addition, we believe that it improves the scientific validity of our systematic review of case reports.

Reviewer Name: Anonymous
Review Date: 2022-02-25 11:00
Specific Comments To Authors: Dear Authors, Your work is very important in every day practice. The article content contributes to the development of knowledge about the RSD but I have some recommendations about the Reference list: You have to change reference 18 in part in Discussion about "Current clinical guidelines"...fixed combination therapy is a first choice in modern hypertension th. (according ESH 2018) You added too old references in this article (ref. 12 and ref. 18), please change them to novel JNC 9 Hypertension guidelines 2021 and ESH 2018 Hypertension guidelines or WHO 2021 hypertension guidelines. Additionally, you have to put all the remaining references in the same order (see ref. 8, 9, 11, 13, 14, 15...).
Authors: Thank you for this great insight on our manuscript. We agree that this is the work on RSD aids in building a greater clinical understanding of this interventional procedure. Regarding your suggestions, our team truly appreciates them. We have updated our citations to NOT use the old JNC 6 guidelines but rather replaced them with the 2018 ESH hypertension guidelines and the WHO 2021 guidelines. This led to use replaces the stages of hypertensions with grades of hypertension as seen in 2018 ESH. In addition, we believe this revision allows for a better understanding of how the grades of hypertension correlate to medication regimens and morbidity risks found in additional literature worldwide. Secondly, we have placed all references in numerically order as seen in the flow of the manuscript and evaluated grammar with a native English speaker. Again, thank you for strengthening our manuscript.